Author’s response to reviews

Title: Dentists’ Attitudes Towards Chairside Medical Conditions Screening in a Dental Setting in Saudi Arabia: An Exploratory Cross-Sectional Study

Authors:
Saba Kassim (saba262003@gmail.com)
Badr Othman (dr.badre.othman@gmail.com)
Sakher AlQahtani (asakher@ksu.edu.sa)
Alemad Mustafa Kawthar (alemad-Kawthar@hotmail.com)
Sterling McPherson (smcpherson05@wsu.edu)
Barbara Greenberg (barbara_greenberg@nymc.edu)

Version: 3 Date: 23 Jul 2019

Author’s response to reviews:

Dentists’ Attitudes Towards Chairside Medical Conditions Screening in a Dental Setting in Saudi Arabia: An Exploratory Cross-Sectional Study

Thank you for accepting our paper. The required amendments have been done.

Editor Comments:
1. Please insert the heading 'Declarations' for the Declarations section, above the 'Acknowledgements' section of your manuscript.
Answer: The heading 'Declarations' has been added above the acknowledgements’ heading Page 11

2. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. Currently Alemad Mustafa Kawthar has been represented using their full name. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
Answer: Alemad Mustafa Kawtharnow replaced with AK Page 11 under Authors’ contribution.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should
remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer: The manuscript has been uploaded as a single file with the Figure and Tables uploaded as separate files.